Author's response to reviews

Title: Associations between quality of life and socioeconomic factors, functional impairments and dissatisfaction with received information and home-care services among survivors living at home two years after stroke onset

Authors:

Michèle Baumann (michele.baumann@uni.lu)
Sophie Coufignal (sophie.coufignal@crp-sante.lu)
Senad Karavdic (senad.karavid@uni.lu)
Etienne Le Bihan (etienne.lebihan@uni.lu)
Nearkasen Chau (nearkasen.chau@wanadoo.fr)

Version: 2  Date: 10 March 2014

Author's response to reviews: see over
Dear Editor,

Thank you very much for your response and the relevant comments of the reviewers for the revised version of our paper entitled:

Associations between quality of life and socioeconomic factors, functional impairments and dissatisfaction with received information and home-care services among survivors living at home two years after stroke onset

Michèle Baumann, Etienne Le Bihan, Kénora Chau, Nearkasen Chau

Please find enclosed the revised version and our responses to the reviewers’ comments. The changes in the manuscript are in red colour.

This manuscript has been seen and approved by all authors, which have been personally and actively involved in substantive work leading to this article, and will hold themselves jointly and individually responsible for its content. The manuscript is not simultaneously under consideration for publication in another journal and the material has not been published elsewhere. All relevant ethical safeguards have been met in relation to subjects’ protection. We have no conflict of interest to declare.

Looking forward to hearing from the Editorial Board,
On behalf of the authors,
Sincerely yours,

Prof Michèle BAUMANN
University of Luxembourg
Research Unit INSIDE
Route de Diekirch BP 2
L7201 - WALFERDANGE
+352 46 66 44 9224
Michele.baumann@uni.lu